Author's response to reviews

Title: Evaluation of the Edinburgh Post Natal Depression Scale using Rasch analysis

Authors:

Julie F Pallant (jpallant@swin.edu.au)
Renee L Miller (r.miller@bigpond.net.au)
Alan Tennant (alantennant@compuserve.com)

Version: 5  Date: 5 June 2006

Author's response to reviews:

Hello BMC,

We have removed the equation number as it is not specifically referred to in the document. We have included the acknowledgements and source of funding. We have cropped the figures to their minimum size. However, this cropping does not show on the upload page under the web layout column, where the graphs are displayed.

I hope this is satisfactory

Kind Regards

Alan Tennant